Author’s response to reviews

Title: Choice of relative or cause-specific approach to cancer survival analysis impacts estimates differentially by cancer type, population, and application: evidence from a Canadian population based cohort study

Authors:

Diana Withrow (diana.withrow@nih.gov)

Jason Pole (jpole@pogo.ca)

E. Diane Nishri (diane.nishri@cancercare.on.ca)

Michael Tjepkema (michael.tjepkema@canada.ca)

Loraine Marrett (loraine.marrett@cancercare.on.ca)

Version: 1 Date: 11 Apr 2017

Author’s response to reviews:

RTR